Author's response to reviews

Title: Evaluating the effectiveness of using personal tailored risk information and taster sessions to increase the uptake of smoking cessation services: study protocol for a randomized controlled trial.

Authors:

Hazel Gilbert (hazel.gilbert@ucl.ac.uk)
Stephen Sutton (srs34@medschl.cam.ac.uk)
Richard Morris (richard.morris@ucl.ac.uk)
Steve Parrott (sdp22@york.ac.uk)
Simon Galton (Simon.Galton@nclondon.nhs.uk)
Irwin Nazareth (i.nazareth@ucl.ac.uk)

Version: 4 Date: 10 October 2012

Author's response to reviews:

Title has been changed to:
Evaluating the effectiveness of using personal tailored risk information and taster sessions to increase the uptake of smoking cessation services: study protocol for a randomized controlled trial.